Author’s response to reviews

Title: Implementing referral guidelines: Lessons from a negative outcome cluster randomised factorial trial in general practice

Authors:

Moyez Jiwa (moyez.jiwa@curtin.edu.au)
Paul Skinner (paul.skinner@sth.nhs.uk)
Akinoso Olujimi Coker (Olujimi.Coker@dbh.nhs.uk)
Lindsey Shaw (lindsey@shaw-home.freeserve.co.uk)
Michael Campbell (M.J.Campbell@sheffield.ac.uk)
Joanne Thompson (J.Thompson1@sheffield.ac.uk)

Version: 5 Date: 31 October 2006

Author’s response to reviews:

We are responding to your request for formatting requirements.

We are attaching 4 figures in word instead of the illegible figure 1. We were unable to produce a legible version of the fifth figure in the old figure 1 and have removed it. Hence only four.

Figure 2 is now figure 5 and has been re-formatted. We note that the scale on our original submission was incorrect and have corrected this error.

The references are cited as per your instructions.

The tables are now part of the manuscript.